Author's response to reviews

**Title:** Antimicrobial susceptibility of travel-related Salmonella enterica serovar Typhi isolates detected in Switzerland (2002-2013) and molecular characterization of quinolone resistant isolates

**Authors:**

Magdalena Nüesch-Inderbinen (ils@fsafety.uzh.ch)
Helga Abgottspon (benzr@fsafety.uzh.ch)
Grethe Sägesser (saegesserg@fsafety.uzh.ch)
Nicole Cernela (n.cernela@access.uzh.ch)
Roger Stephan (stephanr@fsafety.uzh.ch)

**Version:** 2  
**Date:** 28 August 2014

**Author's response to reviews:**

Dear Editorial Team

with this revision we have integrated the requests from the Editorial Team
i) title page: email addresses of all authors, and the full postal address of the submitting author were added
ii) the ethics statement was added: The Zurich Cantonal Ethics Commission evaluated the present study, and declared it legally and ethically unobjectionable.

Kind regards
Roger Stephan